Author's response to reviews

Title: Renal Athersosclerotic reVascularization Evaluation (RAVE Study): Study protocol of a randomized trial [NCT00127738]

Authors:

Sheldon W Tobe (sheldon.tobe@sunnybrook.ca)
Mostafa Atri (mostafa.atri@sunnybrook.ca)
Nancy J Perkins (nancy.perkins@sunnybrook.ca)
Robyn Pugash (robyn.pugash@sunnybrook.ca)
Chaim M Bell (bellc@smh.toronto.on.ca)

Version: 2 Date: 8 June 2006

Author's response to reviews:

Corrected spelling mistakes. Changed titles to sentence format including the tables. Changed references so all authors included and no et als.